Author's response to reviews

Title: Evaluation of an early detection tool for social-emotional and behavioral problems in toddlers: The Brief Infant Toddler Social and Emotional Assessment - A cluster randomized trial

Authors:

Ingrid Kruizinga (i.kruizinga@erasusmc.nl)
Wilma Jansen (w.jansen@ggd.rotterdam.nl)
Alice S Carter (AliceS.Carter@umb.edu)
Hein Raat (h.raat@erasusmc.nl)

Version: 3 Date: 21 June 2011

Author's response to reviews: see over
Concerning  Revision of the manuscript to BMC Public Health “Evaluation of an early detection tool for social-emotional and behavioral problems in toddlers: The Brief Infant Toddler Social and Emotional Assessment – A cluster randomized trial”

Dear Ms. Patel MRCP, PhD,

Please find enclosed a revised version of the manuscript entitled: “Evaluation of an early detection tool for social-emotional and behavioral problems in toddlers: The Brief Infant Toddler Social and Emotional Assessment – A cluster randomized trial”,

The following changes were made:

- Each affiliation is now written in full (Department, Institute, Street, City, Postal code, Country); streets and postal codes were added.
- Subheading ‘author details’ is deleted at the end of the manuscript, since that information (affiliation details) is now provided on the title page.
- The format of all subheadings in the abstract are the same (this is also adjusted in the submission system)

Thank you again for your consideration of our work. We are looking forward to your response.

Sincerely yours on behalf of all authors,

H. Raat MD, PhD
Corresponding author
Erasmus MC – University Medical Centre Rotterdam
Department of Public Health
PO box 2040
3000 CA Rotterdam
The Netherlands
Email: h.raat@erasusmc.nl
Phone: +31 10 4087714
Fax: +31 10 70 38474